A Systematic Review investigating the Effectiveness of Exercise training in Glycogen Storage Diseases

Claire Bordoli, Elaine Murphy, Ian Varley, Graham Sharpe and Philip Hennis

Abstract

Introduction: Glycogen storage diseases (GSDs) are rare inborn errors of carbohydrate metabolism typically with skeletal muscle and liver involvement. In those with skeletal muscle involvement, the majority display symptoms of exercise intolerance which can cause profound exercise limitation and impair everyday living and quality of life (QoL). There are no curative treatments for GSDs, thus therapeutic options, such as exercise training, are aimed at improving QoL by alleviating signs and symptoms. In order to investigate the effectiveness of exercise training in adults with GSDs, we systematically reviewed the literature.

Methods: In this review we conducted searches within SCOPUS and MEDLINE to identify potential papers for inclusion. These papers were independently assessed for inclusion and quality by two authors. We identified 23 studies which included aerobic training, strength training or respiratory muscle training in patients with McArdles (n=41) and Pompe disease (n=139).

Results: In McArdle disease, aerobic exercise training improved aerobic capacity (VO_2 peak) by 14–111% with further benefits to functional capacity and well-being. Meanwhile, strength training increased muscle peak power by 100–151% and reduced disease severity. In Pompe disease, a combination of aerobic and strength training improved VO_2 peak by 9–10%, muscle peak power by 64%, functional capacity and well-being. Furthermore, respiratory muscle training (RMT) improved respiratory muscular strength [maximum inspiratory pressure (MIP) increased by up to 65% and maximum expiratory pressure (MEP) by up to 70%], with additional benefits shown in aerobic capacity, functional capacity and well-being.

Conclusion: This adds to the growing body of evidence which suggests that supervised exercise training is safe and effective in improving aerobic capacity and muscle function in adults with McArdle or Pompe disease. However, the literature base is limited in quality and quantity with a dearth of literature regarding exercise training in other GSD subtypes.

Plain Language Summary

A Systematic Review investigating the Effectiveness of Exercise training in Glycogen Storage Diseases

Background: Glycogen storage diseases (GSDs) are a group of rare inherited metabolic disorders typically affecting carbohydrate metabolism within the skeletal muscle and liver. In those where the skeletal muscle is affected, the majority display exercise intolerance which can impact everyday living and quality of life. No curative treatments for GSDs exist, therefore exercise training offers a therapeutic option to reduce symptoms and enhance patient's quality of life. In order to investigate the effectiveness of exercise training as a therapeutic option, we systematically reviewed the literature.

Systematic Review

Ther Adv Rare Dis

2022, Vol. 3: 1–24 DOI: 10.1177/ 26330040211076497

© The Author(s), 2022. Article reuse guidelines: sagepub.com/journalspermissions

Correspondence to: Claire Bordoli Sport, Health and Performance Enhancement (SHAPE) Research Centre, Nottingham Trent University, Clifton Lane, Clifton, Nottingham NG11 8NS, UK.

claire.bordoli2020@ my.ntu.ac.uk

Elaine Murphy Charles Dent Metabolic Unit, The National Hospital for Neurology and Neurosurgery, London, UK

Ian Varley

Graham Sharpe Philip Hennis Sport, Health and Performance Enhancement (SHAPE) Research Centre, Nottingham Trent University, Nottingham, UK

journals.sagepub.com/home/trd



Creative Commons CC BY: This article is distributed under the terms of the Creative Commons Attribution 4.0 License (http://www.creativecommons.org/licenses/by/4.0/] which permits any use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the SAGE and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

Study Characteristics: We found 23 studies which included aerobic training, strength training and respiratory muscle training in adults with certain types of GSD including McArdles (n = 41) and Pompe disease (n = 139).

Key Results:

- In McArdle disease, aerobic exercise improved aerobic performance, with further benefits to functional capacity and well-being. Meanwhile, strength training increased muscular strength and reduced disease severity.
- In Pompe disease a combination of aerobic and strength training improved aerobic capacity, muscular strength, functional capacity and well-being. Furthermore, respiratory muscle training (RMT) improved the strength of respiratory muscles with further benefits in aerobic capacity, functional capacity and well-being.

Implications: This systematic review indicates that supervised exercise training is safe and effective in improving aerobic capacity and muscle function in adults with McArdle or Pompe disease. However, the effectiveness of exercise training in other GSDs is as yet unknown.

Keywords: endurance training, exercise, glycogen storage disease, respiratory training, review, strength training

Received: 29 October 2021; revised manuscript accepted: 4 January 2022.

Introduction

Glycogen storage diseases (GSDs) are a rare heterogeneous group of inherited disorders of metabolism (IEM) usually caused by single pathogenic variants in genes that encode enzymes involved in carbohydrate metabolism.¹ This results in enzyme deficiencies and subsequent defects of glycogen degradation, glycolysis or paradoxically glycogen synthesis.² Typically, symptoms manifest in skeletal muscle and liver,^{3,4} but the kidneys, heart and in some cases the brain can also be affected.5,6 The specific enzyme deficiency and tissue involvement are used to classify people to one of at least 16 recognised GSD types,⁷ which currently range from Type 0 to Type XV.8 The genetic defects, clinical features and epidemiology of each GSD are summarised by Kanungo et al.5 GSDs manifest along a disease spectrum from being asymptomatic in some patients towards serious pathophysiological implications in others, which can impact peoples' physical health, quality of life (QoL) and life expectancy.5

The physical manifestations of GSDs can become apparent from early childhood to late adulthood, with symptoms and the severity of symptoms varying greatly between different types of GSDs.^{9–11} They are broadly categorised into those with hepatic involvement (GSD0a, GSDIa, GSDIb, GSDVI, GSDIXA1, GSDIXA2, GSDIXc), those with skeletal muscle involvement (GSD0b, GSDII, GSDV, GSDVII, GSDIXD, GSDX, GSDXI, GSDXII, GSDXIII, GSDX1V, GSDXV) and those with both hepatic and skeletal muscle involvement (GSDIII, GSDIV, GSDIXb).⁵ Those with hepatic involvement commonly present with fasting hypoglycaemia, with or without hepatomegaly and liver disease.12 This can influence exercise tolerance due to the direct effects of liver glycogen content on exercise capacity as shown in rodents¹³ and the indirect effect of glycogen via its role in the maintenance of blood glucose homeostasis.14 In contrast where there is skeletal muscle involvement, skeletal myopathy is present.¹² Those with skeletal muscle involvement can typically be divided into those showing static symptoms with loss of muscle mass and strength (GSDII, GSDIII) and those with dynamic exercise-related symptoms of fatigue, muscle pain and contractures, often associated with exercise-induced muscle damage (GSDV, GSDVII, GSDIXD, GSDX, GSDXIV).15 However, clinically these phenotypes can overlap, and precise classification can be challenging.¹⁵ In the GSDs with muscle involvement, exercise intolerance can lead to compromised habitual functioning, with increased morbidity and even premature death in some.¹⁶⁻²⁰ In addition, as a likely

consequence of exercise intolerance, many people with GSDs lead a sedentary lifestyle, which itself is associated with unwanted metabolic adaptations and further health issues.²¹

There are currently no curative treatments available for GSDs, thus therapeutic options are aimed at improving OoL by alleviating signs and symptoms.²² Dietary treatment varies based on the underlying enzyme defect and pathophysiology. Within hepatic GSDs nutritional therapy focusses upon preventing hypoglycaemia although there is a lack of general consensus on the optimal treatment.^{7,8} Enzyme replacement therapy (ERT) is an emerging drug treatment, which has proven benefits in Pompe disease (GSD II) but is not currently available for most GSD subtypes.23 Other supportive measures such as noninvasive ventilation (NIV) and cough assist devices are also employed for respiratory support and airway clearance in late-onset Pompe disease (LOPD) and GSDIII.^{10,24} Although the primary treatments of diet modification are beneficial for GSDs with hepatic and skeletal involvement (particularly in Types 0, I, III, VI, IX and XI) and ERT is relatively successful in reducing symptoms, impairments in functional capacity and OoL still persist.^{7,8} As such, there is a need to identify other treatments to accompany diet and ERT to further improve the health outcomes of people with GSDs. One such intervention is physical exercise training.

Exercise as an intervention may seem counterintuitive to many patients and clinicians given the severe exercise intolerance associated with many GSDs.²⁰ However, increasingly, evidence suggests exercise can be beneficial in reducing symptoms and increasing QoL, rather than accelerating the disease.¹⁵ The three primary exercise interventions considered as treatments for GSDs are aerobic, resistance and respiratory muscle training.

Endurance exercise acts as a powerful inducer of metabolic changes in skeletal muscle. Chronic adaptations associated with training include improvements in substrate delivery to contracting muscles and an increased ability to oxidise non-esterified fatty acids¹⁵ at the same absolute and relative intensity post training.²⁵ For this reason, in McArdle disease, improvements in aerobic capacity and work rate have been found, leading to greater exercise tolerance.^{17,26,27} Theoretically, endurance exercise could potentially have

important benefits to those GSDs with hepatic involvement too, as this shift towards an increased reliance on fat as a fuel and the reduction in plasma glucose oxidation rates would subsequently be protective against hypoglycaemia. Meanwhile, strength training can reverse muscle weakness and atrophy, attenuating disease severity.²⁸ Respiratory muscle training (RMT) is an emerging treatment in Pompe, involving resistance exercise specifically targeting the respiratory muscles, aiming to alleviate significant respiratory weakness.²⁹ These exercise interventions will also combat the physical inactivity seen across the GSD spectrum and in doing so may improve overall general health, fitness and QoL.^{30,31}

Researchers and clinicians have promoted the potential therapeutic benefits of exercise training for people with GSDs for many years. Supervised training programmes have even been included in consensus guidelines for those with Pompe disease.^{32,33} However, despite the beneficial effects of exercise training in GSDs being acknowledged, the research supporting the utility of exercise training in GSDs is sparse and heterogeneous, most likely due to the rarity of these diseases. To date, the only previous systematic review identified three studies and concluded that aerobic exercise effectively induces adaptations in cardiac, metabolic and skeletal muscle activity without adverse events in those with McArdle disease.³⁴ However, no other GSDs were reviewed by Quinlivan et al.34 and since its publication a number of exercise intervention studies have been published, including a randomised controlled trial (RCT).³⁵ As such, several questions remain, which include: Does current literature support the use of exercise training for people with GSD? Which GSD subtypes benefit from exercise training? Which training modalities are effective? And do patients adhere to prescribed exercise interventions? We aim to systematically review the current literature using a defined and reproducible strategy to investigate the broad impact of exercise training programmes across the GSD spectrum and establish the effects that various exercise interventions have on markers of cardiorespiratory and aerobic performance, muscular strength, functional capacity and well-being. In doing so, we aim to determine the feasibility and utility of using exercise training as a treatment option in those with GSDs to inform further research, clinical guidelines and practical recommendations.

Methods

This systematic review is reported in line with the Preferred Reporting Items for Systematic Reviews and Meta-Analyses³⁶ (Additional file 1).

Eligibility

Criteria for inclusion of publications within this review. Eligibility criteria were based on the PICO approach. Inclusion was based on the following criteria:

- A study population with a medical diagnosis of glycogen storage disease, adults (≥18 years).
- Studies comparing the effects of all forms of physical training programmes including aerobic training in the form of swimming, cycling, walking, jogging, flexibility, strength and respiratory muscle training.
- Interventions undertaken for a period of at least 4 weeks to ensure sufficient time for aerobic or respiratory conditioning to occur, but no upper limit on study duration.

Papers were excluded that were unrelated, duplicated, non-human subjects, subject requiring continuous invasive/continuous NIV, unavailable full texts, abstract only papers, editorials, reviews, authors responses and books.

Search strategy

Electronic searches. We searched SCOPUS (1966 to February 2020) using the following search terms: {glycogen storage disease} AND 'exercise' OR 'endurance training' OR ' aerobic exercise' OR 'physical fitness' OR 'muscle training' OR 'respiratory training' OR ' aerobic conditioning' OR 'respiratory training' OR 'walk*' OR 'swim*' OR 'cycl*' OR 'jogging'. Limited to human/humans' papers.

We searched MEDLINE (1950 to February 2020) using a modified search strategy to include exploded MeSH Headings: (MeSH HEADING: exp: (((((((((glycogen storage disease) OR Fanconi Syndrome) OR Glycogen Storage Disease Type I) OR Glycogen Storage Disease) OR Glycogen Storage Disease Type II) OR Glycogen Storage Disease Type III) OR Glycogen Storage Disease Type IV) OR Glycogen Storage Disease Type V) OR Glycogen Storage Disease Type VII) OR Glycogen Storage Disease Type VII) OR Glycogen Storage Disease Type VI) OR Glycogen Storage Disease Type VI) OR Glycogen Storage Disease Type VIII) OR Glycogen Storage Disease Type VIII) OR Glycogen Storage Disease Type VIII) OR HEADING:exp: ((((((exercise) OR Exercise Test) OR Exercise Therapy) OR Exercise) OR Exercise Tolerance) OR Muscle Stretching Exercises) OR Resistance Training)) OR MeSH HEADING:exp: (respiratory training OR Breathing Exercises). All searches were conducted on 11 February 2020.

Selection of studies

Two authors (P.H., C.B.) independently reviewed abstracts in order to identify potential studies for inclusion. Full texts were downloaded and screened for inclusion according to eligibility criteria by two researchers independently. Any disagreement was resolved by consensus agreement following discussion with another author (I.V.). In addition, the reference list of eligible papers was checked to ensure that all potential eligible papers had been identified. Foreign language studies were translated into English.

Quality assessment

Publications were assessed for quality by considering characteristics that could introduce bias using the NIH Quality Assessment Tool for Before-After (Pre-Post) Studies with no control group and the NIH Quality Assessment Tool for controlled intervention studies³⁷ (Additional file 2).

Data extraction

Data from the included studies were extracted into defined tables by a single reviewer. Information was recorded on population characteristics, study design, intervention and outcomes. For Aerobic and Strength training, outcomes were categorised into the following groups: Cardiorespiratory fitness; Muscular strength; Functional capacity and Well-being; and Ventilatory function (where data available). For Respiratory Muscle Training, outcomes were categorised into the following groups: Muscular strength; Ventilatory function; and Functional capacity and Well-being. Data presented in graphs were extracted by Web plot digitiser.³⁸

Results

Search results

A total of 4868 titles and abstracts were screened following the search, 121 of which were included



Figure 1. PRISMA flow diagram of literature screening and selection

for full test screening. Twenty-three articles were subsequently selected for final inclusion in this review (Figure 1). All identified studies included adult patients with McArdle or Pompe disease. Eight studies of adults with McArdle disease were identified, with seven studies investigating aerobic exercise^{17,26,27,39-42} and one study investigating strength training.²⁸ Fifteen studies of adults with Pompe disease were identified, with six investigating a combination of aerobic and muscular training;43-48 two investigating aerobic and nutrition interventions49,50 and seven investigating RMT.35,51-56 No studies including adults with other GSDs were identified. The 23 included publications were largely uncontrolled intervention trials. Other publications included single-arm A-B-A experimental design,53 a double-blind RCT,³⁵ a A-B-C single-arm experimental design,⁵⁶ a longitudinal observation study,55 an uncontrolled prospective study⁵⁰ and a quasi-experimental reversal design study.28

Outcomes reported in patients undergoing aerobic and muscular training included markers of cardiorespiratory fitness, muscular strength and functional capacity and well-being. Markers of cardiorespiratory fitness included maximum work rates, VO₂ peak, submaximal VO₂, ventilatory threshold (VT), gross efficiency (GE), heart rate (HR) and cardiac output. Markers of muscular strength included peak power, Medical Research Council sumscore (MRCss), isometric strength and repetitions of weights. Markers of ventilatory function included vital capacity (VC). Markers of functional capacity included grip strength, timed function tests (TFTs), functional status (Walton & Gardner-Medwin Scale) and muscle function deterioration. All well-being outcomes were selfreported, with validated surveys such as Short Form-36,^{39,42,44,49} Fatigue Severity Scale (FSS)⁴⁴ and Phenotype Severity Scale used.²⁸ In patients undergoing RMT, markers of respiratory muscle strength included maximum inspiratory pressure (MIP), maximum expiratory pressure (MEP) and diaphragm thickness. Markers of ventilatory function included peak cough flow (PCF), forced vital capacity (FVC), forced expiratory volume in 1s (FEV1), MMRC-Dyspnoea and capillary capnometry. Functional capacity and well-being markers included functional status (WGMS),

Author	Quality rating	Population (age)	Study design
Olivier <i>et al.</i> ²⁷	Poor	5 McArdle patients (35 \pm 11 years)	Nonrandomised intervention
Haller et al. ²⁶	Fair	8 McArdle patients (33–61 years)	Nonrandomised uncontrolled intervention
Lucia <i>et al.</i> 40	Poor	1 McArdle and Myasthenia gravis patient (29 years)	Nonrandomised uncontrolled intervention
Mate-Munoz <i>et al.</i> ¹⁷	Fair	10 McArdle patients (36 \pm 5 years)	Nonrandomised intervention
Perez et al. ⁴¹	Poor	1 McArdle patient (38 years)	Nonrandomised uncontrolled intervention
Porcelli <i>et al.</i> ⁴²	Fair	7 McArdle patients (41 \pm 13 years)	Nonrandomised uncontrolled intervention
Cakir <i>et al.</i> ³⁹	Poor	2 McArdle patients (33 years, 42 years)	Nonrandomised uncontrolled intervention
Santalla <i>et al.</i> ²⁸	Fair	7 McArdle patients (23–58 years)	Quasi experimental reversal

 Table 1. McArdle disease: population characteristics and study design.

TFTs, QoL using the Nottingham Health profile (NHP), Quality of sleep using the Pittsburgh Sleep Quality Index (PSQI), reported outcomes of fatigue (FSS) and daytime sleepiness (ESS) and Dyspnoea (SRGQ and MMRC Dyspnoea Scale).

Findings

The following analysis focusses on the effectiveness of exercise in McArdle and Pompe disease. The main exercise interventions identified were aerobic training, muscular training and respiratory muscle training.

Aerobic training in McArdle disease. Characteristics and quality. Aerobic training involving 34 adults with McArdle disease was included in 7 studies, 3 of which were deemed fair quality^{17,26,42} with the remaining 4 studies being deemed of poor quality.^{27,39–41} The basic characteristics of these are shown in Table 1. All of these were nonrandomised intervention studies, with two studies including controls of clinically healthy subjects; however, these were only included for comparison of acute exercise responses at baseline.^{17,27} The duration of training varied from 4 to 32 weeks, with the frequency of training between 3 and 5 times per week and protocols including 60 min or less of walking and running and cycling, with the majority at an intensity of between 60% and 85% max HR.^{17,26,27,39-42}

Outcomes. Cardiorespiratory fitness: Aerobic capacity was shown to improve with increased VO₂ Peak observed in five studies (shown in Table 2).^{17,26,40-42} Greatest improvements were found by Perez et al.⁴¹ (14.6 to 30.8 ml/kg/min), Lucia et al.⁴⁰ (8.5 to 17.0 ml/kg/min) and Mate-Munoz et al.17 $(13.0 \pm 3.8 \text{ to } 18.8 \pm 5.9 \text{ ml/kg/min})$. Smaller differences were seen by Haller et al.26 (1.3 to 1.5 L/ min) and Porcelli *et al.*⁴² (18.5 \pm 1.8 to 21.6 \pm 1.9 ml/ kg/min). In contrast, no differences were found by Olivier et al.²⁷ VT increased in Lucia et al.⁴⁰ (6.1 to 11.8 ml/kg/min) and Mate-Munoz et al.17 $(9.4 \pm 1.8 \text{ to } 12.8 \pm 3.7 \text{ ml/kg/min})$. Gross muscle efficiency was only found to increase by Perez et al.⁴¹ (13.8% to 17.2%). Three studies showed improvements in peak heart rate (Perez et al.41 140 to 166 bpm; Mate-Munoz *et al.*¹⁷ 146 \pm 22 to 156 ± 19 bpm; Lucia *et al.*⁴⁰ 141 to 163 bpm), whereas others found no differences.42 Similar increases were found in cardiac output (Haller et al.26 13.1 to 15.0 L/min; Porcelli et al.42

lable z. Aer	obic and strengt	n training in McA	rdle disease			
Author	Duration	Frequency	Protocol of sessions	Outcomes		
	(weeks)	laays/week)		Cardiorespiratory fitness	Muscular strength	Functional capacity and well-being
Aerobic trai	ning					
Olivier et al. ²⁷	œ	e	Cycling 35–45 min Intensity 60–70% max HR Then 60 min recovery	Submaximal VO₂: No differences HR : Decreased at submaximal workloads	1	1
Haller et al. ²⁶	14	4	Cycling 30–40 min Intensity 60–70% max HR	Peak Work rate Increased 36% [+19 watts] ($p < 0.002$] VO2 Peak during prolonged test: Increased 14% (+0.2 L/min] Peak Cardiac output [Q]: Increased 15% (+2 L/min) ($p < 0.02$]	1	1
Lucia et al. ⁴⁰	12	ы	Walking ≼60 min Intensity 60% max HR	Peak Work rate Increased 61% [+38 watts] VO₂ Peak: Increased 100% [+8.5 ml/kg/min] VT: Increased 93% [+5.7 ml/kg/ min] Peak HR: Increased 16% [+22 bpm]	1	Self-reported improved sense of well-being and ability to perform activities of daily living
Munoz et al. ¹⁷	32	വ	Walking and cycling and gentle running ≼60 min at 60% peak HR	Peak Work rate: Increased 38% (+0.3 W/kg) (p =0.014) V0_2 Peak: Increased 45% (+5.8ml/ kg/min) (p =0.006) VT: Increased 36% (+3.4 ml/kg/ min) (p =0.012) GE: No differences (p =0.476) Peak HR: Increased 7% (+10 bpm) (p =0.05)	I	I
Perez et al. ⁴¹	28	3-4	Running ≼60 min Intensity	Peak Work rate: Increased 103% [+61 watts] VO₂ Peak: Increased 111% (+16.2 mL/kg/min) GE: Increased 3.4% Peak HR: Increased 19% [+26 bpm]	I	Self-reported improvement in well- being and ability to perform activities of daily living
						(Continued)

C Bordoli, E Murphy et al.

Table 2. (Conti	nued)					
Author	Duration	Frequency	Protocol of sessions	Outcomes		
	(weeks)	(days/week)		Cardiorespiratory fitness	Muscular strength	Functional capacity and well-being
Porcelli et al. ⁴²	12	4	1) 10–15 min Stretching exercises 2) 30–45 min cycling 65–70% max HR	Peak Work rate: Increased 22% [+16 watts] [p = 0.02] VO ₂ Peak: Increased 17% [+3.1 ml/ kg/min] [p = 0.02] GE: Increased 0.8% Peak HR: No differences Peak KP: Increased 22% [19.9 ml] [p < 0.05] Peak Cardiac Output [O]: Increased 24% [+3.7 L/min] [p = 0.04]	I	Quality of life: No differences
cakir etal. ³⁹	4	ری	1) Walking 30–45 min Intensity 5–44% 2) 3 reps of static stretching 3) Diaphragmatic breathing also included	1	1	Grip strength: R: Increased 9% (+2.0 kg) L: Increased 14% (+1.8 kg) 10 m walking time: Decreased 15% (-1.08 s) Time to climb 4 steps: Decreased 7% (-0.17 s) Sit to stand within 30 s: Increased 14% (+1.5 s). Ouality of life: Improvement in all scores, except for physical function, vitality, and emotional role in one patient
Strength train	ing					
Santalla et al. ²⁸	16 [+8 detraining]	7	 Warm up (12 min on arm crank ergometer and 12 min on cycle ergometer) Circuits using large muscle groups, 5-6 reps, using load (kg) eliciting RPE of 6-7. Bench press, leg press, lateral pull down, abdominals Passive stretching (3 × 30 s for each muscle group) 	1	Peak Power: Bench press: Increased 100% $\{+52$ watts} (p = 0.018) Half Squat: Increased 151% $\{+173$ watts] (p = 0.018)	All changed to a lower severity class
GE, gross effici	ency; SV, stroke v	olume; VT, ventila	tory threshold; Heart Rate (HR), Rating	g of perceived exertion (RPE).		

Author	Quality rating	Participants with LOPD (age)	Study design
Leutholtz and Ripoll ⁴⁶	Poor	N=1 (24 years)	Nonrandomised uncontrolled intervention trial
Terzis <i>et al.</i> ⁴⁸	Fair	N=5 (36-71 years)	Nonrandomised uncontrolled intervention trial
Van den Berg <i>et al.</i> 43	Good	N=23 patients (46 years; 19.6–70.5)	Nonrandomised uncontrolled intervention trial with staggered starts
Favejee <i>et al.</i> ⁴⁴	Good	N=23 patients (46 years; 19.6–70.5)	Nonrandomised uncontrolled intervention trial
Sechi <i>et al.</i> 49	Good	N=13 (49±11.0years)	Partially blinded, randomised, crossover study
Slonim <i>et al.</i> ⁵⁰	Fair	N=34 (44 ± 11 years)	Uncontrolled prospective study
Montagnese <i>et al.</i> 47	Poor	N=2 (52 and 74 years)	Nonrandomised uncontrolled Intervention trial
Khan <i>et al.</i> ⁴⁵	Poor	N=1 (34 years)	Nonrandomised uncontrolled intervention trial
LOPD, late-onset Pompe d	isease.		

Table 3. Pompe disease: population characteristics and study design.

 15.2 ± 1.3 to 18.9 ± 1.1 L/min). Peak work rate in an incremental test increased in all five studies that measured this^{17,26,40-42} with the highest overall increase being 59 to 120 W.⁴¹ Significant increases were reported by Haller *et al.*²⁶ (59 to 78 W), Mate-Munoz *et al.*¹⁷ (0.8 ± 0.2 to 1.1 ± 0.3 W/kg) and Porcelli *et al.*⁴² (73 ± 13 to 89 ± 12 W).

Functional capacity and well-being: Improvements in functional capacity (grip strength and 10 m walk) were observed by Cakir *et al.*³⁹ (Right grip: 21.5 to 23.5 kg; Left grip: 20.5 to 22.5 kg; 10 m walk: 7.0 to 5.9 s). Improvements in well-being were reported in three studies^{39–41} with two studies documenting self-reported improvements in well-being^{40,41} and others reporting improvements in all QoL scores.³⁹

Adherence: In the only study to report this outcome, Porcelli *et al.*⁴² observed 96% adherence to their training programme.

Strength training in McArdle disease. Characteristics and quality. Strength training was included in one study consisting of seven patients with McArdle disease²⁸ (Table 1). This was a quasi-experimental reversal trial deemed of fair quality, consisting of resistance training including a warmup, circuit training of large muscle groups followed by passive stretching, carried out twice per week for 16 weeks.

Outcomes. Muscular power was shown to increase after resistance training (Table 2; Bench press: 51 to 103 W; Half Squat: 116 to 290 W).²⁸

Adherence: Adherence to training was reported to be between 84% and 100%.²⁸

Aerobic and strength training in Pompe. Characteristics and quality. A combination of aerobic and resistance exercises was investigated in 4 studies consisting of 29 of adults with Pompe disease.^{43,44,46,48} The basic characteristics of these studies are shown in Table 3. These were a mixture of poor,⁴⁶ fair⁴⁸ and good quality studies.^{43,44} Participants were reported to be receiving ERT in three studies^{43,44,48} and hormone replacement therapy in one study.⁴⁶ Two studies included patients supported by walking aids^{43,44} and others included patients supported with NIV.⁴⁶ All of these studies, with Van den Berg *et al.*⁴³ and Favejee *et al.*⁴⁴ reporting on different outcomes from the same study population. Interventions included aerobic and strength^{46,48} and aerobic, strength and core stability exercise.^{43,44} The frequency of training was 3 times per week with a varied duration between 12 and 20 weeks. Aerobic training consisted of 30 min cycling.^{43,44,46,48} Strength training included major muscle groups, using either body weight or weights^{46,48} or exercise machines.^{43,44} In addition, core stability exercises were included in two studies.^{43,44}

Outcomes. Cardiorespiratory fitness: VO₂ peak increased after aerobic and strength training and was broadly consistent with aerobic training alone (Table 4; Van den Berg *et al.*⁴³ 22.1 ± 7.0 to 24.1 ± 7.1 ml/min/kg). Increases were also found in VT (16.7 ± 4.3 to 18.5 ± 4.7 ml/min/kg) and maximum work rate (110 ± 52 W to 122 ± 53 W).⁴³

Ventilatory function: VC was only found to increase by Leutholtz and Ripoll⁴⁶ (1.2 to 1.5 L).

Muscular strength: Improvements in muscular strength were shown in three studies^{43,46,48} with increases in maximum isometric strength found by Terzis et al.⁴⁸ (Hip extension: 4.6 ± 3.3 kg to 6.9 ± 3.7 kg; Bench press: 12.2 ± 5.3 kg to 15.2 ± 7.8 kg, Rowing: 16.7 ± 9.0 kg to 19.8 ± 5.9 kg) using a load transducer and Van den Berg et al.43 (Hip flexors: 156.4 ± 61.9 N to 180.7 ± 57.7 N, Shoulder abductors: 143.1 ± 29.1 N to $150.7 \pm$ 35.4N) using hand-held dynamometry of individual muscle groups. An increase in resistance training repetitions (Curls: 10 to 15; Leg extensions: 7 to 10; Pullovers: 10 to 20; Chest presses: 10 to 20) using a 10-lb weight was observed by Leutholtz and Ripoll.46

Functional capacity and well-being: Functional capacity was shown to improve in two studies^{43,48} with improvements found by Terzis *et al.*⁴⁸ (6MWT: 204 ± 177 m to 248 ± 184 m) and Van den Berg *et al.*⁴³ (6MWT: 492 ± 89 to 508 ± 97 m; Time to climb four steps -0.54 to -0.04 s and supine-stand time: -2.0 to 0.01 s), in contrast to others.⁴⁶ Well-being outcomes included reduced fatigue (median scores: 5.33 to 4.78) and the numbers of patients reporting pain (13/23 versus 5/23).⁴⁴

Adherence: Adherence was only reported in 1 of 3 studies, which found high (89%) rates of session completion.⁴³

Aerobic. strength and nutrition interventions in Pompe. Characteristics and quality. Aerobic, strength and nutrition interventions were described by 2 studies, consisting of 47 adults with LOPD, which were deemed of good⁴⁹ or fair quality⁵⁰ (shown in Table 3). Participants were receiving ERT in the study by Sechi et al.49 with some patients supported by NIV in the study by Slonim et al.⁵⁰ These studies were a partially blinded, randomised crossover study49 and an uncontrolled prospective study.50 There was a large variation in duration from 26 weeks to 10 years, at a frequency of training between 4 and 7 times per week. Aerobic training consisted of 30-40 min cycling (at 11-13 Borg Scale or 60% VO₂ or 60% max HR) or 45–50 min (at 60–65% max HR) on a treadmill, with strength training including 10-15 min using exercise machines or resistance bands. All patients were recommended to consume a caloric distribution of 25-30% protein, 30-35% carbohydrate and 35-40% fat, with the ingestion of l-alanine, 1.5 g, 4 times/day by Slonim et al.⁵⁰ In contrast, patients were randomly assigned to exercise alone or exercise + diet by Sechi et al.⁴⁹ in which the dietary intervention was a personalised high protein diet (25-30% protein, 30-35% carbohydrate, 35-40% fat).

Outcomes. Cardiorespiratory fitness: Significant improvements were reported in peak work rate after aerobic exercise alone (although no difference in median was found 75 ± 80 to 75 ± 70 W). With aerobic exercise and diet there were improvements in peak work rate (median 63 ± 55 to 73 ± 55 W) and VO₂ Peak (median: 22.2 ± 7.3 ml/min/kg to 22.2 ± 4.6 ml/min/kg).⁴⁹

Ventilatory function: VC did not change.49,50

Muscular strength: No differences were found by Sechi *et al.*;⁴⁹ however, there was a significant improvement in muscle function deterioration after nutrition, aerobic and resistance exercise by Slonim *et al.*⁵⁰ [-0.29 (95% CI -0.36, -0.19)].

Functional capacity and well-being: Significant improvements in general health and vitality after aerobic, resistance exercise and nutrition were also found.⁴⁹

Adherence: Twenty-six of the thirty-four patients were considered to have moderate to good compliance to the combined exercise and nutrition

Table 4. Aerobi	ic and muscu	lar interventi	ons in Pompe disease				
Author	Duration	Frequency	Protocol of sessions	Outcomes			
	(weeks)	ladys/week)		Cardiorespiratory fitness	Muscular strength	Ventilatory function	Functional capacity and well-being
Aerobic and stren	gth training						
Leutholtz and Ripoll ⁴⁶	12	ო	Aerobic exercise: 30 min Cycle RPE 11 to 13 of Borg scale 30 min Strength training: 30% of 1RM, 12–15 reps using 'Pyramid' CAM assisted circuit machines	1	Repetitions: [at wt. 10 lbs] Curls: Increased 50% (10–15] Leg ext.: Increased 42% (7–10) Pullovers: Increased 100% (10–20) Chest press: Increased 100% (10–20]	VC: Increased 31% [1.2–1.5 L]	Sit to stand: Unable to complete
Terzis <i>et al.</i> ⁴⁸	50	ო	 30 min cycling 10 min stretching of major muscle groups. Resistance exercises of major muscle groups 1-3 sets of 10 reps, resistance 50% 10 reps max. 	1	Knee extension: $[MVC]$ Increased 46% $(+2.3 \text{ kg})$ $(p > 0.05]$ Hip extension: $[MVC]$ Increased 51% $(+2.3 \text{ kg})$ $(p < 0.05]$. Bench press: $[MVC]$ Increased 25% $(+3.0 \text{ kg})$ $(p < 0.05]$. Rowing: $[MVC]$ Increased 19% $(+3.1 \text{ kg})$ $(p < 0.05]$	1	6MWT: Increased 22% [+44 m] [p < 0.01]
Van den Berg et al. ⁴³	2	m	 5 min Warm-up, intensity 100–110 bpm 15 min Cycling (Intensity 60% VO₂ max) Strength training (Weight 70% of 4 RM, 3 sets 15–20 reps) 15 min Cycling 15 Core stability 5 Core stability 3 sets of 30s) 5 min Cool down 	$\begin{array}{l} \textbf{Peak Work rate:}\\ lncreased\\ 11\% \left\{+12 \text{ watts}\right\}\\ \left\{\rho<0.01\right\}\\ \textbf{VO2} \textbf{peak:} lncreased\\ 9\% \left(2.0 \text{ mU/min/kg}\right)\\ \textbf{V}=0.01\\ \textbf{V}=(0.01)\\ \textbf{V}=(1.1\%)\\ \left\{p<0.01\right\}\\ \left\{p<0.01\right\} \end{array}$	Hip flexors: (MVC) Increased 15% ($+ 24$ N) ($p < 0.01$) Shoulder abductors: (MVC) Increased 5% ($+ 7.6$ N) ($p = 0.02$) Others: No differences	VC: No differences	6MWT: Increased 3% (+16 m) (ρ = 0.01) Time to climb 4 steps: Decrease 12% (-0.3s) (ρ = 0.02) Rise to standing: Rise to standing: (ρ = 0.05) Others: No difference
Favejee et al. ⁴⁴	12	ო	As Van den Berg <i>et al. ⁴³</i>	1	1	1	Fatigue: Decreased 10% (Medians: 5.33 to 4.78) (ρ = 0.007) Pain : Decreased 35% (ρ = 0.0401. Mental Health:

C Bordoli, E Murphy et al.

(Continued)

No differences

Author	Duration	Frequency	Protocol of sessions	Outcomes			
	Imeers	(Naaw/skpn)		Cardiorespiratory fitness	Muscular strength	Ventilatory function	Functional capacity and well-being
Aerobic and strengt	h training plus	dietary interventi	uo				
Sechi <i>et al.</i> ⁴⁹	26	4	Aerobic exercise: 1) Warm up 2) 10–15 min Stretching and balance 3) 15 min Strength, moderate loads of main muscle groups using elastic bands. 3 × 10 reps. 4) 30–40 min Cycling (intensity 60% max HR) Diet: 25–30% protein, 30–35% carbohydrate, 35–40% fat	Peak Work rate: Increased after exercise (medians 75 \pm 80 to 75 \pm 70 watts (p = 0.023) and 16% (+ 10 watts) and 16% (+ 10 watts) and 16% (+ 10 watts) and 16% (+ 10 watts) and 16% (+ 2.023) 3. 53 ± 55 (p = 0.093) 1. 0.2 , 1 ± 2.0 m// min/kg) (p = 0.093) (p = 0.009) after exercise and diet V1 . No differences	Arms extensors, Arm flexors, Leg extensors, Legs flexors: No differences (MVC)	VC: No differences	6MWT: No differences Walton score: No differences. General health: Increased $\{p = 0.03\}$ after exercise and diet Vitality: Increased $\{p = 0.03\}$ after exercise and diet Other components: No differences
Slonim <i>et al.</i> ⁵⁰	104-520 [234 ± 130]	4	Aerobic exercise: 1) 45–50 min treadmill (or cycle) Intensity 60–65% max HR 2) 10–15 min upper body exercise Nutrition: 25–30% protein, 30–35% Carbohydrate 35–40% fat + L-alanine 1.5 g/day 4x/day	1	1	VC: No differences	Difference between pre- NET slope of muscle function deterioration to that of post-NET slope was (-0.29 (95% Cl, 0.19, 0.39) ($p < 0.0001$)
Vibration training							
Montagnese et al. ⁴⁷	104	m	2 × 3 min standing 2 × 30 s semi push up position	1	MRCss: Increased 14% (+6 N) Knee extension: (MVC) Increased 44% (+35.5 bil. Nma) Arm flexion: (MVC) Increased 77% (+29.6 bil. Nma)	VC: No differences	6MWT : Decreased 13% [–19 m] TFTs and WGMS: No differences
Khan <i>et al.</i> ⁴⁵	15	ო	Cycle: 60 s vibration on then 60 s vibration off. 2 cycles initially then progressing to 4. Standing	1	Peak power: increased 64% [+53 watts). Knee extensors: [MVC] Increased 17% [+6 Nm] Flexors: [MVC] Decreased 13% [-2 Nm]	VC: No differences	6MWT: Increased 70% (+116 m) Mean grip: No differences
CI, confidence int Maximal Voluntar	erval; VC, vita y Muscle Con	ıl capacity; VT, traction; RM, F	/entilatory threshold; WGMS, well-be tepetition Maximum; 6MWT, Six Minu	eing markers included fu ute Walk Test; CAM, Com	nctional status; HR, Heart Rate; RI outer aided manufacturing.	PE, Rating of per	ceived exertion; MVC,

Therapeutic Advances in Rare Disease 3

Table 4. (Continued)

Author	Quality rating	Participants with LOPD (age)	Study design
Martin <i>et al.</i> ⁵⁴	Poor	N=1 (42 years)	Nonrandomised uncontrolled intervention trial
Mitja <i>et al.</i> ⁵⁵	Fair	N=8 (13-58years)	Longitudinal observational study
Aslan <i>et al.</i> ⁵¹	Fair	N=8 (23-64 years)	Nonrandomised uncontrolled intervention trial
Wenninger <i>et al.</i> ⁵⁶	Fair	N=11 (50 ± 15.6 years)	Prospective monocentric unblinded single-arm pilot study A-B-C design
Jones <i>et al.</i> ⁵²	Poor	N=2 (55 and 64years)	Nonrandomised uncontrolled intervention trial
Jones <i>et al.⁵³</i>	Fair	N=8 (49.3 ± 8.4 years)	A-B-A single subject experimental design
Jones <i>et al.</i> ³⁵	Fair to Good	N = 22 RMT: 12 patients (53.2 ± 12.7 years) Sham-RMT:10 patients (46.6 ± 13.9 years)	Double-blind randomised control trial
LOPD, late-onset Pom	npe disease; RMT, re	spiratory muscle training.	

Table 5. Pompe disease: population characteristics and study design.

intervention.⁵⁰ Similarly, Sechi *et al.*⁴⁹ found reasonable adherence to exercise (69% for warm up, 61% for strength training, 74% for cycling, 69% for stretching) with no significant difference between arms. Within the dietary intervention, the percentage of calories introduced with proteins was 96% (median value) of those prescribed and significantly increased from habitual diet.

Strength training in Pompe disease. Characteristics and quality. Whole body vibration training (WBVT) and side alternating vibration training (SAVT) were described by two nonrandomised uncontrolled intervention studies which were deemed of poor quality^{45,47} and included three adults with LOPD (shown in Table 3). Both studies included patients supported by walking aids, with some receiving ERT and additional regular physiotherapy.⁴⁷ Training was carried out 3 times per week; however, there was a large variation in duration of training ranging from 15 to 104 weeks.^{45,47}

Outcomes. Improvements in muscular strength included increases in MRCss (41.5 to 47.5), knee extension (70.8 to 106.3 Nm), arm flexion (38.3 to 67.9 Nm)⁴⁷ and peak power using jumping

mechanography (83 to 136 W).⁴⁵ Improvements in functional capacity were also found (6MWT: 166 to 282 m)⁴⁵ (shown in Table 4).

Adherence: This was not reported in either study.

Respiratory interventions in Pompe disease. Characteristics and quality. Respiratory muscle training was described in a total of 7 studies consisting of 60 adults with LOPD (shown in Table 5).^{35,51–56} Most of the studies were deemed fair quality,^{51,53,55,56} with only Jones *et al.*³⁵ deemed fair to good quality.

The population characteristics and study design are shown in Table 5. All included patients had respiratory muscle weakness, with five papers including a subset of patients receiving noninvasive nocturnal ventilation (including between 1 and 13 patients).^{35,51,53,54,56} Adjuvant ERT treatment was used in all but one study.⁵⁴ Three studies were nonrandomised uncontrolled trials.^{51,52,54} Jones *et al.*³⁵ was the only double-blind RCT. Others included an A-B-A single subject trial,⁵³ a single-arm pilot study A-B-C design⁵⁶ and a longitudinal observational study.⁵⁵ A summary of the interventions and outcomes are shown in Table 6. All of the papers

	Functional capacity and well-being			Gardner-Medwin- Walton scale: No differences	Quality of life: Social isolation scores: Improved (Medians: 2.5 ($22.1-69$ km 0.0 ($0.0-16.9$) ($p=0.02$) Other Subscores: No differences Sleep Quality: No differences	6MWT, quality of life (SGRQ, MMRC-Dysnea scale): No differences
	Ventilatory function		FEV ₁ : No differences FVC: Decreased 7% [-0.32]	FVC: No differences	FVC, FEV1, PCF: No differences	FVC, FEV1, Capillary capnometry: No differences
Outcomes	Muscular strength		MIP: Increased 45% [+27.0cmH20] MEP: Increased 70% [+42.0cmH20]	MIP: Increased 18% $(+5.6 \text{ cm}\text{H}_2\text{O})$ $(p=0008)$ Improvements stable over the course of the study $(p < 0.05)$ MEP. No differences	MIP: Increased 30% $[+9.0 \text{ cm H}_2\text{O}]$ $[p = 0.01]$ MEP: No differences	MIP: Increased 16% (+7.6 cmH ₂ O) (<i>p</i> =0.024) Increased 26**% (+13.4 cmH ₂ O)*** (<i>p</i> =0.001) MEP: No differences
Protocol of	sessions		15 min x 2/day.	Cycle: 1' at 30% MIP then 2' deep slow breathing 15 cycles per day for 45 min (15' at 30% MIP and 30' at rest with deep slow breathing]	15 min, twice/ day 80 sessions per patient	30 min daily 7 × 15 inhalations each 525 IMT reps per week
Intensity	(% MIP OF MEP)		6-46 cmH ₂ 0 L/sec	30	30 initially, increased weekly by 2 cmH20	30–40 initially then optional increase by 10–15.
Frequency	(aays/ week)		7	2	¶ ∭	പ
Duration (weeks)		cle training	15	96	ω	6 [+6 detraining + 40 optional training period]
Author		Inspiratory muso	Martin et al. ⁵⁴	Mitja e <i>t al.</i> 55	Aslan <i>et al.</i> 51	Wenninger et al. ⁵⁶

(Continued)

Table 6. Respiratory muscle training in Pompe disease..

Table 6. (Continu	led)						
Author	Duration (weeks)	Frequency	Intensity	Protocol of	Outcomes		
		ldays/ week)	(% MIP OF MEP)	sessions	Muscular strength	Ventilatory function	Functional capacity and well-being
Inspiratory and	expiratory muscle trai	ining					
Jones <i>et al.</i> ⁵²	16-32	\$	≥60	2 × 25 reps of IMT or EMT daily for 4–10weeks then both IMT and EMT	MIP: Increased 65% (+18.0 cmH ₂ 0) MEP: Increased 39% (+17.0 cmH ₂ 0)	FVC: No differences*	
Jones <i>et al.</i> ⁵³	12	ى	60-70	3 × 25 reps of IMT and EMT daily Overall: 4500 reps IMT 4500 reps EMT	MIP: Increased 20% $[+8.6 \text{ cmH}_2\text{O}]$ MEP: Increased 16% $[+11.6 \text{ cmH}_2\text{O}]$	PCF: Increased 12% [+1.0 L/s]****	6MWT: Increased 2.2% (+7m) Supine to stand: Decreased 13% (-1.6s) Stair climbing: Decreased 15% (-0.6s) 10 m walk: Decreased 3% (-0.3s)
Jones <i>et al.</i> ³⁵	12	ى	RMT: 50–70 Sham-RMT: 15	3 × 25 reps of IMT and EMT daily. Overall: 4500 reps EMT 4500 reps EMT	MIP: No differences (between groups) MEP: No differences (between groups) Diaphragm Thickness: No differences	PCF: No differences Polysomnography: no differences	Time to climb 4 steps: Decreased 0.9 s in RMT group. Decreased 0.1 s in Sham-RMT group ($p = 0.0346$) Daytime steepiness (ESS): Decreased 1.2 in RMT group. Increased 1.1 in sham-RMT ($p = 0.0160$) (no raw data available) Other gross motor function and patient- reported outcomes: No differences (between groups)
EMT, expiratory m expiratory pressu Questionnaire; - N training; ***Over t	nuscle training; ESS, day re; MIP, maximum inspir Vot included; * Obtained the total study period of t	time sleepiness atory pressure; 10 weeks after i 52 weeks includ	; FEV1, forced ex PCF, peak cough discontinuation o ling 6 weeks of tr	piratory volume in 1s; F n flow; RMT, respiratory f RMT in 1 patient and a aining, 6 weeks detraini	-VC, forced vital capacity; IMT · muscle training; 6MWT, Six 1 ifter approximately 6 months ing and a 40 week optional tra	, inspiratory muscle traini Vinute Walk Test; SGRQ, S of RMT in the second patic aining period; **** Only m	ing; MEP, maximum 5t. Georges Respiratory ent; **After 6 weeks of neasured in 5 patients.

included RMT, with both Inspiratory Muscle Training (IMT) and Expiratory muscle training (EMT) used in three studies^{35,52,53} and IMT used alone in four studies.^{51,54–56}

The majority of the studies had a training period of 15 weeks or less, with two papers of longer training duration of up to 96 weeks.^{52,55} The interventions used pressure threshold RMT which consisted of multiple repetitions of RMT against fixed resistance (threshold) at either a relatively fixed intensity^{35,52,53,55} or progressive intensities over the duration of the training.^{51,54,56} Patients were prescribed between 30 and 45 min training per day, between 5 and 7 times per week.

Outcomes. Respiratory muscle strength: Improvements in markers of muscular strength were found in six of the seven studies after IMT alone or with both IMT and EMT (Table 6). Significant improvements in MIP were observed in three studies, specifically after IMT [Mitja et al.⁵⁵ 31.6 \pm 17.7 to $37.2 \pm 19.3 \text{ cmH}_2\text{O}$; Aslan et al.,⁵¹ median $30 \text{ cmH}_2\text{O}$ (21.5–48.0) to $39 \text{ cmH}_2\text{O}$ (31.2–56.5); Wenninger *et al.*⁵⁶ 48.6 ± 18.0 to $56.2 \pm$ 19.9 cmH₂O after initial 6-week training period and 48.6 ± 18.0 to $61.4 \pm 28.7 \text{ cmH}_2\text{O}$ over the total duration of study including an additional training period]. Improvements in MEP were observed in three studies; however, in contrast, these were predominantly observed after inspiratory and expiratory training combined and values showing statistical significance were not presented.52-54

Ventilatory function: Improvements in ventilatory function were only found by Jones *et al.*⁵³ with increases in PCF (7.5 ± 0.6 to 8.5 ± 1.4 L/s). No differences were found in other ventilatory markers.

Functional capacity and well-being: Improvements in markers of functional capacity were found after IMT and EMT [Jones *et al.*:⁵³ 6MWT: Increased 2.2% (+6.9 m); Supine-stand: Decreased 13.4% (-1.6 s), climb 4 stairs: Decreased 15% (-0.6 s), walk 10 m: Decreased 3.4% (-0.3 s) Jones *et al.*:³⁵ Climb 4 steps -0.9 s treatment *versus* -0.1 s control, p=0.03]. Well-being including social isolation scores significantly improved [Aslan *et al.*⁵¹ Median 22.5 (22.1–69.8) to 0.0 (0.0–16.9)] and daytime sleepiness significantly decreased compared to controls (Jones *et al.*³⁵ ESS score -1.2 treatment *versus* +1.1 controls). Adherence: Adherence was excellent, in which Wenninger *et al.*⁵⁶ found 107% of the training sessions were completed. Similarly, Jones *et al.*⁵³ found a mean of 99% prescribed IMT repetitions and 101% of prescribed EMT repetitions were completed and Jones *et al.*³⁵ found mean adherence to be 98% in the treatment arm and 97% in the control arm.

Discussion

The aim of this review was to investigate the effectiveness of exercise training in adults with GSDs. Of the recognised 16 forms of GSDs,⁷ no data were available for 14, with exercise interventions only assessed in adults with McArdle or Pompe disease. In McArdle disease, aerobic exercise in five of the seven studies improved aerobic performance,17,26,40-42 with further benefits on functional capacity and well-being found by Lucia et al.40 and Perez et al.41 Strength training increased muscular strength and reduced disease severity²⁸ (see Table 2) and these improvements were found to be retained in all but one patient after a period of 2 months detraining.²⁸ In Pompe disease, the literature shows that a combination of aerobic and strength training improves aerobic capacity, muscular strength, functional capacity and well-being (see Table 4). Furthermore, RMT improved respiratory muscular strength in all of the studies of Pompe disease, with additional benefits in aerobic capacity, functional capacity and well-being shown by some (see Table 6). In the current literature, exercise training appears to be safe and beneficial to health in adults with McArdle and Pompe disease and thus there appears to be a growing body of evidence which suggests that supervised exercise training is safe and effective in improving aerobic capacity and muscle function in adults with McArdle or Pompe disease. The literature base is limited, however, in both quality and quantity with a dearth of literature regarding exercise training in other GSD subtypes. Further research of robust design would be beneficial across the spectrum of GSDs.

Aerobic and strength training in McArdle disease

McArdle disease (OMIM #232600, EC 2.4.1.1) is caused by a deficiency of myophosphorylase (PYGM), which impairs glycogenolysis specifically in skeletal muscle.^{5,28,57} This block in glycogenolysis limits glucose availability (via glycogen) and places greater reliance on blood glucose and non-esterified fatty acids to meet the metabolic needs of skeletal muscle. During exercise, when the metabolic rate is elevated, the impairment in substrate availability can result in an acute energy crisis and result in pain, fatigue and contractures.^{15,57} In acute exercise of moderate intensity, these symptoms can subside as exercise duration persists beyond 10min due to greater uptake of blood-borne glucose derived from the liver and fat oxidation in contracting muscles.^{28,57-59} However, despite some improvement in exercise capacity during moderate intensity, during longer duration exercise people with McArdle disease still exhibit impaired exercise performance and aerobic capacity compared to a healthy population.^{15,28,57} Due to this, the majority of studies in McArdle disease investigated the effect of aerobic training, which is hypothesised to increase exercise tolerance primarily through an improvement in aerobic capacity that is driven by a training-induced improvement in free fatty acid oxidation (the 'second wind' effect). In addition, aerobic training may also counteract the effects of a sedentary lifestyle, which can include an increased dependence of glvcogen as a fuel, skeletal muscle atrophy and weakness, and poor cardiovascular fitness.¹⁵

The current literature supports the hypothesis that aerobic training improves exercise capacity in McArdle disease and provides evidence for the mechanisms involved in the improvement. Exercise capacity improved after aerobic training in adults with McArdle disease with increases in peak work rates reported in all studies that measured this outcome (n=5).^{17,26,40–42} The magnitude of exercise performance gains was not only consistent but appeared substantial, with intervention studies reporting mean improvements in maximal cycling work rates ranging between 22% and 38%17,26,42 while case studies reported improvements between 61% and 103%.40,41 Although the improvements documented above are promising, the limited study quality hinders our ability to form strong conclusions and the effect of exercise training on exercise capacity is yet to be fully elucidated. It is likely that the greater exercise capacity following training is partly attributable to improvements in oxidative metabolism, with studies finding concurrent increases in VO₂ peak, indicative of an increase in cardiac output^{17,26,40-42} and increased VT.40 Improvements were even found in an individual with severe clinical features and the presence of two neuromuscular diseases.⁴⁰

The magnitude of this training-induced change is in line with those found in a healthy population, indicating a similar relative capability to improve exercise performance and aerobic capacity, albeit from lower baseline values.⁶⁰

The literature indicates that exercise training in McArdle disease elicits physiological adaptations at several steps of the oxygen cascade, which underpin the improvement in exercise and aerobic capacity, most notably improved cardiovascular function, and muscle oxidative metabolism. Cardiovascular function was markedly improved by training, with 15-24% increases in peak cardiac output^{26,42} achieved via improvements in both peak HR^{17,40,41} and peak stroke volume (SV).42 Furthermore, lower heart rates were recorded during submaximal exercise, also indicating improved SV.27,42 Although the changes in cardiovascular function are considerable and certainly beneficial to overall health,⁶¹ they are likely to facilitate the training-induced improvements in aerobic and exercise capacity, rather than drive them. The primary training adaptation in McArdle disease patients is most likely related to improvements in skeletal muscle oxidative metabolism. In the most in-depth study, Porcelli et al. 42 found that 12 weeks of cycling training at 65-70% max HR resulted in a reduction in the O₂ cost of submaximal exercise $(15.8 \pm 1.3 \text{ to } 13.6 \pm 1.2 \text{ ml/})$ min/W, p = 0.03), faster pulmonary VO₂ kinetics in those with slow VO₂ kinetics before training and lower submaximal RER. These results indicate that skeletal oxidative metabolism is more efficient following training and that substrate utilisation during submaximal exercise is shifted towards greater free fatty acid utilisation. This is further supported by evidence of an increase in the mitochondrial enzymes citrate synthase and B-hydroxyacyl coenzyme A dehydrogenase.^{17,26} However, these metabolic adaptations to exercise training were not uniform across studies, with Olivier et al.27 reporting no difference in sub max VO₂ which is likely attributable to less total training sessions (24 sessions versus ≥ 60 sessions by others) and Mate-Munoz et al.17 reporting no differences in GE, potentially due to exercising at a lower intensity compared to others (Table 2). Due to the improvements in cardiorespiratory fitness shown, it is unsurprising that there were improvements in functional capacity³⁴ and the ability to perform activities of daily living^{40,41} and OoL.³⁹ Surprisingly, despite improvements in outcomes of cardiorespiratory fitness (Peak work rate, VO₂ Peak, GE, Peak SV and Peak cardiac output), Porcelli *et al.*⁴² found no improvements in habitual levels of physical activity or QoL. However, the post training measurements of physical activity and QoL were taken up to 3 months after the termination of training, thus if any improvements were made, they were not found to be sustained.

Strength training improved strength due, in part, to increases in total and lower body lean mass; these improvements appeared to reduce disease severity. However, these results are from just one small study, thus limiting the validity of findings to the wider population.²⁸ Overall, based on the literature reviewed, exercise may be an effective method of reducing symptoms in McArdle disease through improvements in aerobic and physical capacity.

Aerobic and strength training in Pompe disease

Pompe disease (OMIM #232300, EC 3.2.1.20) is a GSD that also affects skeletal muscle; however, in contrast to McArdle disease, the heterozygous mutation in the GAA gene is within the lysosomes and lysosomal glycogenolysis is blocked. As there is minimal contribution from lysosomal glycogen breakdown to ATP production, it is therefore not a deficiency in ATP resynthesis.¹⁸ Instead, the deficiency of alpha-1,4-glucosidase causes an accumulation of lysosomal glycogen in skeletal, respiratory and cardiac muscle which leads to exercise intolerance and skeletal muscle weakness and wasting.5 A sedentary lifestyle further impacts exercise intolerance, with immobility causing skeletal muscle atrophy, weakness with a low VO2 and increased dependence of glycogen as fuel.¹⁵ As Pompe disease primarily impacts muscle weakness, wasting and physical activity, all studies in adults with Pompe disease focussed on muscular interventions alone or in combination with aerobic training.

The literature shows that a combination of aerobic and strength training increased muscular strength, with improvements observed in three of the four studies which measured this.^{43,46,48} The studies that employed formal weight training protocols with clearly defined levels of resistance were able to increase strength by 5–100% (Table 4); however, these were nonrandomised uncontrolled studies including 23 patients or less who exercised between 12 and 20 weeks.^{43,46,48} The trial which showed no improvement in strength employed resistance band training and a subjective marker of intensity, which may account for the lack of improvement.⁴⁹ Following improvements in muscle strength, subsequent increases in functional capacity were found, with increases in 6MWT,43,48 time to climb 4 steps and rise to standing.43 The mechanisms underpinning the strength gains were not investigated in depth; however, increased lean body mass (LBM) accompanied strength gains,48 indicating participants may have undergone training-induced muscle hypertrophy. Indeed, the magnitude of the change in LBM found by Terzis et al.⁴⁸ (+8.4%) is similar to that observed in a healthy population undergoing a similar strength training programme (+8.5%).⁶² These results are encouraging, particularly as the quality of evidence appears greater compared to that of McArdle disease due to the inclusion of large sample sizes and inferential statistics.

Despite only two studies reporting on aerobic performance, significant improvements in maximum work rate,^{43,49} VO₂ peak and VT⁴³ were found indicating that exercise training in Pompe disease elicits physiological adaptations along the oxygen cascade as observed in McArdle disease. Furthermore, the addition of a high protein diet appeared to elicit greater improvements in maximum work rate and an increase VO₂ peak.⁴⁹ Further to these improvements, reductions in fatigue and pain were also found.44 Where aerobic, resistance exercise and diet were combined in an intervention lasting 2-10 years, a reduction in muscle function deterioration was observed, most likely being due to a reduction of glycogen and a reduction in proteolysis, autophagy and muscle damage.^{50,63} Furthermore, Sechi et al.⁴⁹ found improvements in general health and vitality, which is unsurprising following the improvements in aerobic capacity.

Vibration training was shown to improve muscular strength,^{45,47} which could be due to muscle adaptations including type two myofiber hypertrophy and an increased muscle cross-sectional area^{64,65} and may explain improvements in 6MWT by Khan *et al.*⁴⁵ Vibration training could therefore offer a time efficient and easily adopted mode of exercise;^{45,47} however, as results were derived from poor-quality case reports,^{45,47} larger studies of enhanced quality are necessary before recommendations can be made. In summary, all training modalities appear to benefit muscular strength and functional capacity, with improvements in aerobic performance where aerobic exercise is included.

Respiratory interventions in Pompe disease

Progressive respiratory weakness is prevalent in LOPD⁵⁶ with symptoms such as nocturnal hypoventilation, diaphragm weakness or sleep apnoea becoming apparent before other muscle weakness.^{66–68} Despite ERT, respiratory weakness persists in approximately one third of patients, in which respiratory muscle weakness can decline by approximately 3.2% MIP per year.^{69,70} This leads to reduced airway clearance,⁷¹ sleep disordered breathing and the requirement of ventilatory support towards the latter stages. In 70% of patients, this can even progress to premature death.⁷²

RMT was conducted in adults with LOPD as it offers a therapeutic option using pressure threshold respiratory trainers calibrated to provide inspiratory or expiratory resistance for forced voluntary inspiration/expiration muscle contractions.⁵² As inspiratory muscles are similar to skeletal muscle, they should respond to training and enhance ventilation with increased coordination, endurance and strength.⁵⁶ IMT protocols were therefore conducted in all studies, aiming to target specific inspiratory muscle weakness.⁷³

The literature shows that IMT increased inspiratory muscle strength, with MIP improving in six of seven studies.^{51–56} Improvements were observed after just 6 weeks (+15.7% MIP),⁵⁶ increased over the duration of training⁵² and improvements were found to be maintained following training cessation.35,53,56 The addition of EMT to IMT also led to increased expiratory muscle strength in the form of MEP.52,53 In the only randomised double-blind controlled trial by Jones et al.,35 despite the magnitude of improvements in MIP and MEP being similar to others,^{52,53} no statistical differences were found between RMT and Sham-RMT arms due to the control arm appearing to elicit an active response. This study was also underpowered and despite randomisation there were differences in baseline characteristics between groups, with those assigned the treatment being older, on ERT for longer, and with increased respiratory muscle involvement. Interestingly, it appears EMT is necessary to elicit improvements in functional capacity^{35,53} which may be due to the role of respiratory muscles in aiding truncal mobility and stabilisation.^{73–75} The evidence appears to be of fair to good quality due to well-defined participant selection, repeat outcome measures and the inclusion of statistics.

Isolating the effect of RMT on the outcomes reported is difficult as all studies included patients receiving ERT which is known to stabilise or even reduce respiratory decline.⁷⁶ However, improvements from ERT largely occur 12–26 weeks after treatment and are usually modest (3–4% for MIP and MEP).^{23,76} Patients had been receiving ERT for at least 10 months in the majority of studies prior to RMT; thus, the improvements shown were likely attributable to RMT alone.^{35,52,53,55,56}

Safety and adherence. Aerobic exercise in McArdle disease was shown to be well tolerated in the two studies that reported this,^{26,27} with no adverse effects such as muscle injury, contractures, rhabdomyolysis or myoglobinuria. Furthermore, aerobic exercise appeared safe in five studies in which the muscle damage marker creatine kinase (CK) remained stable^{26,27} or even decreased throughout the trial.^{17,40,41} Adherence was only measured in one of the seven studies but was reported to be very good with 96% of training sessions completed, which was most likely due to the high levels of support provided, with weekly phone calls and encouragement given.⁴² Similarly, strength training in McArdle disease was well tolerated, with no adverse effects reported and CK remaining stable. In addition, patients were very complaint, with 100% of sessions completed in five of the seven participants.28

Aerobic and strength training in Pompe disease was reported to be well tolerated in the only study that reported on this, with no adverse events observed⁴⁸ and safe with CK levels shown to decrease over the duration of the trial in the only other study that reported on this.43 Adherence was reported in four of the six studies;43,44,49,50 however, only two studies provided data.49,50 Sechi et al.49 found adherence to exercise was good (61-74%) and Slonim et al.⁵⁰ found 26 out of 34 patients were consistently compliant with the nutrition and exercise protocol. Similarly, both studies, including vibration training in Pompe, found it to be well tolerated, with no adverse events reported and CK remaining stable. However, no measures of adherence were reported.45,47

Respiratory interventions for Pompe were well tolerated with no adverse effects reported in three of the seven studies that reported this.^{51,53,56} Where side effects were reported, the majority were mild and almost half were unrelated.³⁵ Adherence was reported by three of the seven studies and of these it was reported to be excellent, with 107%,⁵⁶ 99% (IMT sessions) and 101% (EMT sessions)⁵³ and 98% (treatment arm) and 97% (control arm) completion rates.³⁵ Wenninger *et al.*⁵⁶ found adherence to decrease over the full duration of the study, with a drop of 13% (107 to 94%) training sessions completed over the total duration which included 6 weeks of detraining and then a 40-week optional training period.

Overall, aerobic, strength and respiratory muscle training appears to be safe and well tolerated with good adherence in McArdle and Pompe patients. However, this was only reported on in a limited number of studies and should be important factors to include in future studies.

Strengths and limitations. The major strength of this review is it is the first to systematically investigate the effectiveness of exercise training across the full spectrum of GSDs, using detailed and reproducible methodology with strict inclusion and exclusion criteria. Considering the full spectrum of GSDs allowed us to identify that exercise training interventions have only been studied in McArdle and Pompe disease, leaving 14 of the 16 GSD types unstudied. We also considered the effectiveness of several exercise training modalities, allowing us to consider the relative benefits of each for both GSD types.

The major limitation of this review was a lack of research into other GSDs, particularly others with skeletal muscle involvement (GSDVII, IXd, X) and those with skeletal muscle and liver involvement (GSD III, GSDXIV), in which the effects of exercise on the skeletal and hepatic presentation could have been investigated.5,19,77 Furthermore, the potential impact of exercise on overall general health, fitness and QoL could have been investigated in non-muscular forms of GSDs (GSD0a, GSDIa, GSDIb, GSDVI, GSDIXA1, GSDIXA2, GSDIXc).^{30,31} The majority of the studies included were deemed poor to fair quality, largely due to being uncontrolled, of small sample sizes including several case studies and of short duration. This deficit in study quality impacts interpretation; however, some limitations

are to be expected given the rarity of these diseases and the impact this has on recruitment. Regarding study design, it is important to acknowledge that there could have been a learning effect of the pre and post intervention tasks which may have impacted on the results reported. In addition, interpretation of the literature was also impaired by the heterogeneity of training interventions, which varied substantially in regard to modality, duration, intensity and the inclusion of other treatment strategies (e.g. ERT). In those receiving ERT, they had been on this for at least 1 year; therefore, the initial benefits of ERT would have stabilised and thus improvements observed were likely due to the interventions alone.43,44,47,49 Furthermore, staggered starts of studies showed patients remained relatively stable and exercise training and nutrition still reduced muscle deterioration in those not receiving ERT.⁵⁰

Further research. Further RCTs and intervention studies should include adults with other GSDs particularly those with skeletal muscle involvement as part of larger multicentre randomised studies to clarify the effectiveness, safety and adherence of exercise training across the broad spectrum of GSDs. However, it is acknowledged this would be difficult given the rarity of these diseases. Comparison between individual exercise components within studies would offer greater insights into the most effective methods of training, with further research into vibration training and RMT warranted given the success in Pompe. Further studies of longer duration with multiple follow-up periods would allow us to see if beneficial effects would be maintained long term and if exercise training has wider implications on preventing chronic diseases such as diabetes and cardiorespiratory disease.²⁷

Conclusion

In conclusion, the current evidence shows exercise training appears to be safe and effective in adults with McArdle disease or LOPD, with improvements observed in aerobic capacity, muscular strength and functional capacity. The effect of RMT in Pompe, where sufficiently intense, was also found to be beneficial, with these improvements appearing to be maintained several months after training stopped. However, these findings are largely based on the limited quality of evidence available, largely derived from small uncontrolled intervention studies of short duration that included highly varied exercise protocols which limits the generalisation of findings. Further research of increased quality is required, particularly focussing on how exercise may benefit the clinical course of the disease across the broad spectrum of GSD types.

Author contributions

Claire Bordoli: Project administration; Writing – original draft; Writing – review & editing.

Elaine Murphy: Writing – review & editing.

Ian Varley: Supervision; Writing – review & editing.

Graham Sharpe: Supervision; Writing – review & editing.

Philip Hennis: Supervision; Writing – review & editing.

Conflict of interest statement

The authors declared no potential conflicts of interest with respect to the research, authorship and/or publication of this article.

Funding

The authors received no financial support for the research, authorship and/or publication of this article.

Supplemental material

Supplemental material for this article is available online.

Ethical approval/patient consent

Ethical approval and informed consent were not required for this systematic review.

References

- Hicks J, Wartchow E and Mierau G. Glycogen storage diseases: a brief review and update on clinical features, genetic abnormalities, pathologic features, and treatment. *Ultrastruct Pathol* 2010; 35: 183–196.
- 2. Smit G, Rake J, Akman H, *et al.* The glycogen storage diseases and related disorders. In: Fernandes J, Saudubray JM, van den Berghe G, *et al.* (eds) *Inborn metabolic diseases: diagnosis and treatment.* Berlin: Springer, 2006, pp. 101–119.
- 3. DiMauro S, Hays A and Tsujino S. Metabolic disorders affecting muscle. In: Engel AG and

Franzini-Armstrong C (eds) *Myology*. New York: McGraw-Hill, 2004, pp. 1535–1558.

- Laforet P, Weinstein D and Smit P. The glycogen storage diseases and related disorders. In: Saudubray JM, Van den Berghe G and Walter JH (eds) *Inborn metabolic diseases*. Berlin: Springer, 2012, pp. 115–139.
- Kanungo S, Wells K, Tribett T, et al. Glycogen metabolism and glycogen storage disorders. Ann Transl Med 2018; 6: 474.
- 6. Aoun B, Sanjad S, Deghelli J, *et al.* Kidney and metabolic phenotypes in glycogen storage disease type-1 patients. *Front Pediatr* 2020; 8: 591.
- Heller S, Worona L and Consuelo A. Nutritional therapy for glycogen storage diseases. *J Pediatr Gastroenterol Nutr* 2008; 47(Suppl. 1): S15–S21.
- 8. Ross KM, Ferrecchia IA, Dahlberg KR, *et al.* Dietary management of the glycogen storage diseases: evolution of treatment and ongoing controversies. *Adv Nutr* 2020; 11: 439–446.
- Bhengu L, Davidson A, du Toit P, et al. Diagnosis and management of Pompe disease. S Afr Med 2014; 104: 273–274.
- Chien Y-H, Hwu W-L and Lee NC. Pompe disease: early diagnosis and early treatment make a difference. *Pediatr Neonatol* 2013; 54: 219–227.
- Schoser B, Stewart A, Kanters S, et al. Survival and long-term outcomes in late-onset Pompe disease following alglucosidase alfa treatment: a systematic review and meta-analysis. J Neurol 2017; 264: 621–630.
- Tarnopolsky MA. Myopathies related to glycogen metabolism disorders. *Neurotherapeutics* 2018; 15: 915–927.
- Lopez-Soldado I, Guinovart JJ and Duran J. Increased liver glycogen levels enhance exercise capacity in mice. *J Biol Chem* 2021; 297: 100976.
- Gozalez JT, Fuchs CJ, Betts JA, et al. Liver glycogen metabolism during and after prolonged endurance-type exercise. Am J Physiol Endocrinol Metab 2016; 311: E543–E553.
- Preisler N, Haller RG and Vissing J. Exercise in muscle glycogen storage diseases. J Inherit Metab Dis 2015; 38: 551–563.
- Haller RG and Lewis SF. Glucose induced exertional fatigue in muscle phosphofructokinase deficiency. N Engl J Med 1991; 324: 364–369.
- 17. Mate-Munoz JL, Moran M, Perez M, *et al.* Favourable responses to acute and chronic

exercise in McArdle patients. *Clin J Sport Med* 2007; 17: 297–303.

- Preisler N, Laforet P, Madsen KL, et al. Fat and carbohydrate metabolism during exercise in late-onset Pompe disease. *Mol Genet Metab* 2012; 107: 462–468.
- Preisler N, Pradel A, Husu E, *et al.* Exercise intolerance in glycogen storage disease type III: weakness or energy deficiency? *Mol Genet Metab* 2013; 109: 14–20.
- 20. Vissing J. Exercise training in metabolic myopathies. *Rev Neurol* 2016; 172: 559–565.
- Stein TP and Wade CE. Metabolic consequences of muscle disuse atrophy. *J Nutr* 2005; 135: 1824S–1828S.
- 22. Stone WL, Basit H and Adil A. *Glycogen storage disease*. Treasure Island, FL: StatPearls Publishing, 2021.
- Van der Ploeg AT, Clemens PR, Corzo D, et al. A randomized study of alglucosidase alfa in lateonset Pompe's disease. N Engl J Med 2010; 362: 1396–1406.
- 24. Gaeta M, Musumeci O, Mondello S, et al. Clinical and pathophysiological clues of respiratory dysfunction in late-onset Pompe disease. New insights from a comparative study by MRI and respiratory function assessment. *Neuromuscul Disord* 2015; 25: 852–858.
- Achten J and Jeukendrup AE. Optimising fat oxidation through exercise and diet. *Nutrition* 2004; 20: 716–727.
- Haller RG, Wyrick P, Taivassalo T, et al. Aerobic conditioning: an effective therapy in McArdle's disease. Ann Neurol 2006; 59: 922–928.
- Olivier K, Hogrel J-Y, Gomez-Merino M, et al. Effects of an endurance training on patients with McArdle's disease. Sci Sports 2005; 20: 21–26.
- Santalla A, Mungui-Izquierdo D, Brea-Alejo L, et al. Feasibility of resistance training in adult McArdle patients: clinical outcomes and muscle strength and mass benefits. *Front Aging Neurosci* 2014; 6: 334.
- Gozal D and Thiriet P. Respiratory muscle training in neuromuscular disease: long-term effects on strength and load perception. *Med Sci Sports Exerc* 1999; 31: 1522–1527.
- Blair SN, Kohl HW, Barlow CE, et al. Changes in physical fitness and all-cause mortality. A prospective study of healthy and unhealthy men. *JAMA* 1995; 273: 1093–1098.

- Pedersen BK and Saltin B. Evidence for prescribing exercise as therapy in chronic disease. *Scand J Med Sci Sports* 2006; 16(Suppl. 1): 3–63.
- Kishnani PS, Steiner RD, Bali D, et al. Pompe disease diagnosis and management guideline. *Genet Med* 2006; 8: 267–288.
- Cupler EJ, Berger KI, Leshner RT, et al. Consensus treatment recommendations for late onset Pompe disease. *Muscle Nerve* 2012; 45: 319–333.
- Quinlivan R, Vissing J, Hilton-Jones D, et al. Physical training for McArdle disease. Cochrane Database Syst Rev 2011; 12: CD007931.
- Jones HN, Kuchibhatla M, Crisp KD, et al. Respiratory muscle training in late-onset Pompe disease: results of a sham-controlled clinical trial. *Neuromuscul Disord* 2020; 30: 904–914.
- 36. Page MJ, McKenzie JE and Bossuyt PM. The PRISMA 2020 statement. An updated guideline for reporting systematic reviews. *BMJ* 2021; 372: n71.
- NHLBI. Study quality assessment tools, https:// www.nhlbi.nih.gov/health-topics/study-qualityassessment-tools (2021, accessed 11 March 2021).
- Rohatgi A. Web Plot Digitizer, 4.4. https:// automeris.io/WebPlotDigitizer (2020, accessed 30 March 2021).
- Cakir T, Filiz M, Otag B, et al. The effect of aerobic exercise with low intensity and short duration on functional capacity and quality of life in two patients with McArdle disease. Turkiye Fiziksel Tip Ve Rehabilitasyon Dergisi 2017; 63: 82–87.
- Lucia A, Mate-Munoz JL, Perez M, et al. Double trouble (McArdle's disease and myasthenia gravis): how can exercise help? *Muscle Nerve* 2007; 35: 125–128.
- 41. Perez M, Moran M, Cardona C, *et al.* Can patients with McArdle's disease run? *Br J Sports Med* 2007; 41: 53–54.
- Porcelli S, Marzorati M, Morandi L, *et al.* Homebased aerobic exercise training improves skeletal muscle oxidative metabolism in patients with metabolic myopathies. *J Appl Physiol* 2016; 121: 699–708.
- 43. Van den Berg LEM, Favejee MM, Wens SCA, *et al.* Safety and efficacy of exercise training in adults with Pompe disease: evaluation of endurance, muscle strength and core stability

before and after a 12-week training program. *Orphanet J Rare Dis* 2015; 10: 87.

- 44. Favejee MM, van den Berg LE, Kruijshaar ME, *et al.* Exercise training in adults with Pompe disease: the effects on pain, fatigue, and functioning. *Arch Phys Med Rehabil* 2015; 96: 817–822.
- 45. Khan A, Ramage B, Robu I, *et al.* Sidealternating vibration training improves muscle performance in a patient with late-onset Pompe disease. *Case Rep Med* 2009; 2009: 741087.
- Leutholtz BC and Ripoll I. The effects of exercise on a patient with severe acid maltase deficiency. *Eur J Phys Med Rehabil* 1996; 6: 185–187.
- Montagnese F, Thiele S, Wenninger S, et al. Long-term whole-body vibration training in two late-onset Pompe disease patients. *Neurol Sci* 2016; 37: 1357–1360.
- 48. Terzis G, Dimopoulos F, Papadimas GK, et al. Effect of aerobic and resistance exercise training on late-onset Pompe disease patients receiving enzyme replacement therapy. *Mol Genet Metab* 2011; 104: 279–283.
- Sechi A, Zuccarelli L, Grassi B, et al. Exercise training alone or in combination with highprotein diet in patients with late onset Pompe disease: results of a cross over study. Orphanet J Rare Dis 2020; 15: 143.
- Slonim AE, Bulone L, Goldberg T, et al. Modification of the natural history of adult-onset acid maltase deficiency by nutrition and exercise therapy. *Muscle Nerve* 2007; 35: 70–77.
- 51. Aslan GK, Gurses HN, Issever H, *et al.* Effects of respiratory muscle training on pulmonary functions in patients with slowly progressive neuromuscular disease: a randomized controlled trial. *Clin Rehabil* 2014; 28: 573–581.
- 52. Jones HN, Moss T, Edwards L, et al. Increased inspiratory and expiratory muscle strength following respiratory muscle strength training (RMST) in two patients with late-onset Pompe disease. *Mol Genet Metab* 2011; 104: 417–420.
- Jones HN, Crisp K, Robey RR, et al. Respiratory muscle training (RMT) in late-onset Pompe disease (LOPD): effects of training and detraining. *Mol Genet Metab* 2016; 117: 120–128.
- Martin RJ, Sufit RL, Ringel SP, et al. Respiratory improvement by muscle training in adult-onset acid maltase deficiency. *Muscle Nerve* 1983; 6: 201–203.
- 55. Jevnikar M, Kodric M, Cantarutti F, *et al.* Respiratory muscle training with enzyme

replacement therapy improves muscle strength in late-onset Pompe disease. *Mol Genet Metab Rep* 2015; 5: 67–71.

- 56. Wenninger S, Greckl E, Babacici H, et al. Safety and efficacy of short- and long-term inspiratory muscle training in late-onset Pompe disease (LOPD): a pilot study. J Neurol 2019; 266: 133–147.
- 57. Scalco RS, Lucia A, Santalla A, et al. Data from the European registry for patients with McArdle disease and other muscle glycogenoses (EUROMAC). Orphanet J Rare Dis 2020; 15: 330.
- Vissing J and Haller RG. The effect of oral sucrose on exercise tolerance in patients with McArdle's disease. N Engl J Med 2003; 349: 2503–2509.
- 59. Ørngreen MC, Jeppesen TD, Andersen ST, *et al.* Fat metabolism during exercise in patients with McArdle disease. *Neurology* 2009; 72: 718–724.
- DeBusk RF, Stenestrand U, Sheehan M, et al. Training effects of long versus short bouts of exercise in healthy subjects. Am J Cardiol 1990; 65: 1010–1013.
- 61. Laukkanen JA, Lakka TA, Rauamaa R, *et al.* Cardiovascular fitness as a predictor of mortality in men. *Arch Intern Med* 2001; 161: 825–831.
- Narici MV, Roi GS, Landoni L, et al. Changes in force cross sectional area and neural activation during strength training and detraining of the human quadriceps. Eur J Appl Physiol Occup Physiol 1989; 59: 310–319.
- 63. Marek M, Jarý J, Valentová O, *et al.* Immobilization of glycoenzymes by means of their glycosidic components. *Biotechnol Lett* 1983; 5: 653–658.
- 64. Schoser B. Muscle histology changes after short term vibration training in healthy controls. *Acta Myol* 2015; 34: 133–138.
- 65. Łochyński D, Kaczmarek D, Rędowicz MJ, et al. Long-term effects of whole-body vibration on motor unit contractile function and myosin heavy chain composition in the rat medial gastrocnemius. J Musculoskelet Neuronal Interact 2013; 13: 430–441.
- Mellies U, Stehling F, Dohna-Schwake C, et al. Respiratory failure in Pompe disease: treatment with non-invasive ventilation. *Neurology* 2005; 64: 1465–1467.
- Mellies U, Ragette R, Schwake C, et al. Sleepdisordered breathing and respiratory failure in acid maltase deficiency. *Neurology* 2011; 57: 1290–1295.

- Margolis ML, Howlett P, Goldberg R, et al. Obstructive sleep apnea syndrome in acid maltase deficiency. *Chest* 1994; 105: 947–949.
- 69. van der Beek NAME, van Capelle CI, van der Velden-van Etten KI, *et al.* Rate of progression and predictive factors for pulmonary outcome in children and adults with Pompe disease. *Mol Genet Metab* 2011; 104: 129–136.
- 70. Regnery C, Kornblum C, Hanisch F, et al. 36 months observational clinical study of 38 adult Pompe disease patients under alglucosidase alfa enzyme replacement therapy. *J Inherit Metab Dis* 2012; 35: 837–845.
- 71. Pitts T, Bordelon R, Huff A, *et al.* Cough effectiveness and pulmonary hygiene practices in patients with Pompe disease. *Lung* 2019; 197: 1–8.

Visit SAGE journals online journals.sagepub.com/ home/trd

SAGE journals

1-8.
72. Boentert M, Karabul N, Wenninger S, *et al.* Sleep-related symptoms, and sleep-disordered breathing in adult Pompe disease. *Eur J Neurol*

2015; 22: 369-376.e27.

- 73. Jung DE, Kim K and Lee SK. Comparison of muscle activities using a pressure biofeedback unit during abdominal muscle training performed by normal adults in the standing and supine positions. *J Phys Ther Sci* 2014; 26: 191–193.
- Lee AY, Kim EH, Cho WY, et al. Effects of abdominal hollowing during stair climbing on the activations of local trunk stabilizing muscles: a crosssectional study. Ann Rehabil Med 2013; 37: 804–813.
- 75. Alejaldre A, Diaz-Manera J, Ravaglia S, et al. Trunk muscle involvement in late onset Pompe disease: study of thirty patients. *Neuromuscul Disord* 2012; 22(Suppl. 2): S148–S154.
- 76. Bembi B, Pisa FE, Confalonieri M, et al. Longterm observational, non-randomized study of enzyme replacement therapy in late-onset glycogenosis type II. J Inherit Metab Dis 2010; 33: 727–735.
- Tegtmeyer LC, Rust S, van Scherpenzeel M, et al. Multiple phenotypes in phosphoglucomutase 1 deficiency. N Engl J Med 2014; 370: 533–542.